Author’s response to reviews

Title: A retrospective analysis of adrenal crisis in steroid-dependent patients: causes, frequency and outcomes

Authors:
Katherine Gwenda White (katherine.white@uni-wuerzburg.de; katherinegwhite@ntlworld.com)

Version: 2 Date: 08 Nov 2019

Author’s response to reviews:

FORMATTING CHANGES:

1. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. For example, the initials of John Smith would be ‘JS’. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors, for example, by adding their middle name initial.

Done

I hope I have understood what you wanted here?